Author's response to reviews

Title: Pulmonary fibrosis secondary to siderosis causing symptomatic respiratory disease: a case report

Authors:

Liam M McCormick (drliammccormick@hotmail.com)
Martin Goddard (Martin.Goddard@papworth.nhs.uk)
Ravi Mahadeva (rm232@cam.ac.uk)

Version: 7 Date: 2 July 2008

Author's response to reviews: see over
To the Editors of the Journal of Medical Case Reports,

Many thanks for considering our case report for submission. In response to the language corrections and formatting, we have made the following adjustments:

**Adjustment 1**

Reference 5. Full list of authors names provided rather than “et al”.

**Adjustment 2**

Short titles for Figures 2 and 3 have been changed to “Figure 2 – Histological Analysis” and “Figure 3 – Histology – Iron and Collagen Stains”. These changes have been made both on the manuscript and on the images themselves.

We hope that the paper is now suitable for publication.

Many thanks,

Liam McCormick, Ravi Mahadeva

Department of Respiratory Medicine,
Addenbrooke’s Hospital,
Cambridge, UK